Author's response to reviews

Title: Mesenteric panniculitis of the sigmoid colon. A case report and review of the literature

Authors:

Angel I Popkharitov M.D;PhD (popkharitov@abv.bg)
Georgi N Chomov M.D. (g_shomov@abv.bg)

Version: 2 Date: 7 March 2007

Author's response to reviews: see over
1. Competing interests list:

My co-author and I declare that we have not competing interests. In the past five years we have not received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future.

My co-author and I declare that we don't hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future.

My co-author and I declare that we don't hold or we are not currently applying for any patents relating to the content of the manuscript. We have not received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript.

We have not any other financial competing interests in relation to this manuscript.

2. Author’s contribution list:

My co-author Dr Georgi Chomov has a substantial contribution of the histological section of the study. He performed the photos and its interpretation.

I, Dr Angel Popkharitov participated in the sequence alignment, explored all possible sources for the references and performed the definitive version of the manuscript.

3. Acknowledgement list:

This work was supported in part by the Thracian University, Medical Faculty, University hospital and District hospital, Stara Zagora, Bulgaria.

Written consent was obtained from the patient for publication of this study.

Dr Angel Popkharitov